Author’s response to reviews

Title: Hugging Stents - misdeployment of two selfexpanding stents into the internal carotid artery: successful interdisciplinary management

Authors:

Thomas Hupp (t.hupp@katharinenhospital.de)
Hellwig Meissner (h.meissner@katharinenhospital.de)
Henning von Loewesprung (h.lowenspring@katharinenhospital.de)
Thomas Guethe (t.guethe@katharinenhospital.de)
Hans Henkes (h.henkes@katharinenhospital.de)
Dominik Jost (jostdominik@gmx.de)

Version: 2 Date: 19 May 2010

Author’s response to reviews:

due to an upload of an earlier version, i have to correct the Abstract page.

last sentence in Conclusion a should be: Further evaluation in larger series is highly recommended.

Please erase the two words: This should...

thank you for your kind of considerations in this matter.